Author's response to reviews

Title: Dietary exposure to methyl mercury and PCB and the associations with semen parameters among Swedish fishermen

Authors:

Anna Rignell-Hydbom (Anna.Rignell-Hydbom@med.lu.se)
Anna Axmon (Anna.Axmon@med.lu.se)
Thomas Lundh (Thomas.Lundh@med.lu.se)
Bo A Jonsson (Bo_A.Jonsson@med.lu.se)
Tarmo Tiido (Tarmo.Tiado@med.lu.se)
Marcello Spano (spanomrc@casaccia.enea.it)

Version: 4 Date: 7 May 2007

Author's response to reviews:

To BioMed Editorial Production Team
Following formatting changes have been done as requested. Line numbers are removed, author list is changed, section headings are changed, footnote is incorporated into text.
Best regards
Anna Rignell-Hydbom